Author's response to reviews

Title: An association between seeing incidents of alcohol or drug use in films and young Scottish adults’ own alcohol or drug use: cross sectional study

Authors:

Kate Hunt (kate@sphsu.mrc.ac.uk)
Helen Sweeting (helen@sphsu.mrc.ac.uk)
James Sargent (James.D.Sargent@Hitchcock.ORG)
Heather Lewars (h_lewars@hotmail.com)
Robert Young (robert@sphsu.mrc.ac.uk)
Patrick West (patbwest@gmail.com)

Version: 4 Date: 27 March 2011

Author's response to reviews:

Thank you for your email of 18.3.2011. We have added the following text to the methods in response to your final query:

Participants / Cohort members were invited to take part via letters with information sheets detailing the survey procedures. Prior to participation they signed a consent form confirming that they had read the information sheet, had the study explained to them and understood what it involved, that the information they would provide was confidential and would be identifiable only by an ID number and that they could choose not to answer any questions they wished.

We hope that you will now find the manuscript acceptable for publication.

Very best wishes, Kate Hunt